Author’s response to reviews

Title: CPMCGLM: An R package for p-value adjustment when looking for an optimal transformation of a single explanatory variable in generalized linear models

Authors:

Benoit Liquet (benoit.liquet@univ-pau.fr)

Jérémie Riou (jeremie.riou@univ-angers.fr)

Version: 1 Date: 28 Feb 2019

Author’s response to reviews:

The responses to the reviewers have been added in the supplementary material section, to provide a more appropriate answer, when an answer based on mathematical proofs is needed.